\draftcut \section{Glossary of notation} \label{sec:glossary} Greek letters, then Latin, then symbols:

\noindent
{\small \begin{multicols}{2}
\begin{list}{}{
  \renewcommand{\makelabel}[1]{#1\hfil}
}

% \alpha
% \beta
% \gamma
% \delta
\item[{$\delta$}] Satoh's tube map\hfill\ \ref{subsec:TangleTopology}
\item[{$\Delta$}] co-product\hfill\ \ref{subsec:ATSpaces}
%\item[{$\Delta$}] cloning\hfill\ \ref{subsec:Quandles}
% \epsilon
% \zeta
% \eta
% \theta
% \iota
\item[{$\iota$}] inclusion $\attr_n\to\calP^w(\uparrow_n)$\hfill\ 
  \ref{subsec:ATSpaces}
% \kappa
% \lambda
% \mu
% \nu
\item[{$\nu$}] the invariant of the unknot\hfill\ \ref{subsec:KTG}
% \xi
% \omicron
% \pi
\item[{$\pi$}] the projection $\calP^w(\uparrow_n)\to\fraka_n\oplus\tder_n$
  \hfill\ \ref{subsec:ATSpaces}
% \rho
% \sigma
% \tau
% \upsilon
% \phi
\item[{$\phi$}] log of an associator\hfill\ \ref{subsec:KTG}
\item[{$\Phi$}] an associator\hfill\ \ref{subsec:KTG}
% \chi
% \psi
\item[{$\psi_\beta$}] ``operations''\hfill\ \ref{subsec:AlgebraicStructures}
% \omega

\item
%a
\item[{$\fraka_n$}] $n$-dimensional Abelian Lie algebra\hfill\ 
  \ref{subsec:ATSpaces}
\item[{$\calA$}] a candidate associated graded structure\hfill\ \ref{subsec:Expansions}
\item[{$\calA^{sv}$}] $\calD^v$ mod 6T, RI\hfill\ \ref{subsec:vw-tangles}
\item[{$\calA^{sw}$}] $\calD^w$ mod $\aft$, TC, RI\hfill\ 
  \ref{subsec:vw-tangles}
\item[{$\calA^{sw}$}] $\grad\wTF$\hfill\ \ref{subsec:fgrad}
\item[{$\calA^{sw}$}] $\grad\wTFo$\hfill\ \ref{subsec:OriFoams}
\item[{$\calA^{(s)w}$}] $\calA^{w}$ and/or $\calA^{sw}$\hfill\ 
  \ref{subsec:fgrad}
\item[{$\calA^u$}] chord diagrams mod rels for KTGs\hfill\ \ref{subsec:KTG}
\item[{$\calA^v$}] $\calD^v$ mod 6T\hfill\ \ref{subsec:vw-tangles}
\item[{$\calA^w$}] $\calD^w$ mod $\aft$, TC\hfill\ \ref{subsec:vw-tangles}
\item[{$\calA^w$}] $\grad\wTFo$ without RI\hfill\ \ref{subsec:fgrad}
\item[{$\calA^-(\uparrow_n)$}] $\calA^-$ for pure $n$-tangles\hfill\ 
  \ref{subsec:ATSpaces}
\item[{$A_e$}] 1D orientation reversal\hfill\ \ref{subsubsec:wops}
\item[{$\Ass$}] associative words\hfill\ \ref{subsec:ATSpaces}
\item[{$\Ass^+$}] non-empty associative words\hfill\ \ref{subsec:ATSpaces}
%b
\item[{$\calB^w_n$}] $n$-coloured unitrivalent arrow \newline
  diagrams\hfill\ \ref{subsec:ATSpaces}
%c
\item[{$C$}] the invariant of a cap\hfill\ \ref{subsec:wTFExpansion}
\item[{CP}] the Cap-Pull relation\hfill\ \ref{subsubsec:wrels},
  \ref{subsec:fgrad}
\item[{CW}] Cap-Wen relations\hfill\ \ref{subsubsec:wrels}
\item[{$c$}] a chord in $\calA^u$\hfill\ \ref{subsec:KTG}
%d
\item[{$\der$}] Lie-algebra derivations\hfill\ \ref{subsec:ATSpaces}
\item[{$\calD^v$, $\calD^w$}] arrow diagrams for v/w-tangles\hfill\
  \ref{subsec:vw-tangles}
\item[{$\divop$}] the ``divergence''\hfill\ \ref{subsec:ATSpaces}
%e
%f
\item[{$F$}] a map $\calA^w\to\calA^w$\hfill\ \ref{subsec:fgrad}
\item[{$F$}] the main~\cite{AlekseevTorossian:KashiwaraVergne} unknown
  \hfill\ \ref{subsec:EqWithAT}
\item[{FR}] Flip Relations\hfill\ \ref{subsubsec:wrels},
  \ref{subsec:fgrad}
\item[{$\fil$}] a filtered structure\hfill\ \ref{subsec:Expansions}
%g
%h
%i
\item[{$\calI$}] augmentation ideal\hfill\ \ref{subsec:Grad}
%j
\item[{$J$}] a map $\TAut_n\to\exp(\attr_n)$\hfill\ \ref{subsec:ATSpaces}
\item[{$j$}] a map $\TAut_n\to\attr_n$\hfill\ \ref{subsec:ATSpaces}
%k
\item[{KTG}] Knotted Trivalent Graphs\hfill\ \ref{subsec:KTG}
%l
\item[{$\lie_n$}] free Lie algebra\hfill\ \ref{subsec:ATSpaces}
\item[{$l$}] a map $\tder_n\to\calP^w(\uparrow_n)$\hfill\ \ref{subsec:ATSpaces}
%m
%n
%o
\item[{$\calO$}] an ``algebraic structure''\hfill\ 
  \ref{subsec:AlgebraicStructures}
%p
\item[{$\calP^w_n$}] primitives of $\calB^w_n$\hfill\ \ref{subsec:ATSpaces}
\item[{$\calP^-(\uparrow_n)$}] primitives of $\calA^-(\uparrow_n)$\hfill\ 
  \ref{subsec:ATSpaces}
\item[{$\grad$}] associated graded structure \hfill\ \ref{subsec:Grad}
%q
%r
\item[{$R$}] the invariant of a crossing\hfill\ \ref{subsec:wTFExpansion}
\item[{R4}] a Reidemeister move for
  \newline foams/graphs\hfill\ \ref{subsubsec:wrels}
%s
\item[{$\sder$}] special derivations\hfill\ \ref{subsec:sder}
\item[{$\calS$}] the circuit algebra of skeletons\hfill\ 
  \ref{subsec:CircuitAlgebras}
\item[{$\SAut_n$}] the group $\exp(\sder_n)$\hfill\ \ref{subsec:KTG}
\item[{$S_k$}] complete orientation reversal\hfill\ 
  \ref{subsec:UniquenessForTangles}
\item[{$S_e$}] complete orientation reversal\hfill\ \ref{subsubsec:wops}
\item[{$\sKTG$}] signed long KTGs\hfill\ \ref{subsec:KTG}
%t
\item[{TV}] Twisted Vertex relations\hfill\ \ref{subsubsec:wrels}
\item[{$\tder$}] tangential derivations\hfill\ \ref{subsec:ATSpaces}
\item[{$\attr_n$}] cyclic words\hfill\ \ref{subsec:ATSpaces}
\item[{$\attr^s_n$}] cyclic words mod degree 1\hfill\ \ref{subsec:ATSpaces}
\item[{$\TAut_n$}] the group $\exp(\tder_n)$\hfill\ \ref{subsec:ATSpaces}
%u
\item[{$u$}] a map $\tder_n\to\calP^w(\uparrow_n)$\hfill\ \ref{subsec:ATSpaces}
\item[{$u_e$}] strand unzips\hfill\ \ref{subsubsec:wops}
\item[{$\uT$}] u-tangles\hfill\ \ref{subsec:sder}
%v
\item[{$V$, $V^+$}] the invariant of a (positive) vertex\hfill\ 
  \ref{subsec:wTFExpansion}
\item[{$V^-$}] the invariant of a negative vertex\hfill\  
  \ref{subsec:wTFExpansion}
\item[{VI}] Vertex Invariance\hfill\ \ref{subsec:fgrad}
\item[{$\vT$}] v-tangles\hfill\ \ref{subsec:vw-tangles}
%\item[{$\vT(\uparrow_n)$}] pure $n$-component v-tangles\hfill\ 
%  \ref{subsec:ATSpaces}
%w
\item[{$W$}] $Z(w)$\hfill\ \ref{subsec:wTFExpansion}
\item[{$W^2$}] Wen squared\hfill\ \ref{subsubsec:wrels}
\item[{$w$}] the wen\hfill\ \ref{subsubsec:wTFgens}
\item[$-$] wenjugation \hfill \ref{subsec:OriFoams}
\item[{$\wT$}] w-tangles\hfill\ \ref{subsec:vw-tangles}
%\item[{$\wT(\uparrow_n)$}] pure $n$-component w-tangles\hfill\ 
%  \ref{subsec:ATSpaces}
\item[{$\wTF$}] w-tangled foams with wens\hfill\ \ref{subsec:wTF}
\item[{$\wTFo$}] orientable w-tangled foams\hfill\ \ref{subsec:OriFoams}
%x
%y
%z
\item[{$Z$}] expansions \hfill\ throughout
\item[{$Z_\calA$}] an $\calA$-expansion\hfill\ \ref{subsec:Expansions}

\item
\item[{4T}] $4T$ relations\hfill\ \ref{subsec:KTG}
\item[{$\uparrow$}] a ``long'' strand\hfill\ throughout
\item[{$\up$}] the quandle operation\hfill\ 
  \ref{subsec:AlgebraicStructures} %\ref{subsec:Quandles}
%\item[{$\up_2$}] doubled $\up$\hfill\ \ref{subsec:Quandles}
\item[{$*$}] the adjoint on $\calA^w(\uparrow_n)$\hfill\ \ref{subsec:ATSpaces}

\end{list}
\end{multicols}}
